Author's response to reviews

Title: Skin reaction to inhaled tiotropium bromide: a case report.

Authors:

Cristoforo Incorvaia (cristoforo.incorvaia@gmail.com)
Nicola Fuiano (fuiano50@tin.it)
Raffaella Megali (raffaella.megali@icp.mi.it)
Gian Galeazzo Riario-Sforza (griariosforza@icp.mi.it)

Version: 2 Date: 17 December 2010

Author's response to reviews:

Dear Editor,
We resubmit the manuscript, revised according to the comment from Reviewer 2.

Comment from Reviewer 2
The possibility of side effects to be produced by excipients of inhaled tiotropium
given the skin prick test was negative was not discussed.

We described more clearly the double-blind, placebo-controlled procedure, that
used as placebo the excipient of inhaled tiotropium, and discussed the negative
result of the skin prick test.